Author’s response to reviews

Title: Associations of AMP and adenosine induced dyspnea sensation to large and small airways dysfunction in asthma

Authors:

Claire Cox (c.a.cox@umcg.nl)
Ilse Boudewijn (i.m.boudewijn@umcg.nl)
Sebastiaan Vroegop (j.s.vroegop@gmail.com)
Siebrig Schokker (s.schokker@umcg.nl)
Anne Lexmond (a.j.lexmond@rug.nl)
Henderik Frijlink (h.w.frijlink@rug.nl)
Paul Hagedoorn (p.hagedoorn@rug.nl)
Judith Vonk (j.m.vonk@umcg.nl)
Martijn Farenhorst (m.p.farenhorst@umcg.nl)
Nick ten Hacken (n.h.t.ten.hacken@umcg.nl)
Huib Kerstjens (h.a.m.kerstjens@umcg.nl)
Maarten van den Berge (m.van.den.berge@umcg.nl)

Version: 2 Date: 03 Jan 2019

Author’s response to reviews:

The manuscript is now composed conform the authors guidelines.

• The Competing interests section is extended with the funders information.

• The Funding section now contains information on the influence the supporting party had on all parts of the research, including its publication.

• Initials have been used in the competing interest section rather than full names.

• An Additional files list is added to the document according to the submission guidelines, mentioning the attachment with supplementary tables.
• The currently uploaded main document, as well as the supplement, is a 'clean', without any tracked changes, comments, highlights, strikethroughs or text in different colors. All tables are situated in the text at logical positions and designed conform the authors guidelines.